Title: Supervised physical exercise to improve the quality of life of cancer patients: the EFICANCER randomised controlled trial

Authors:

Aintzane Sancho (AINTZANE.SANCHOGUTIERREZ@osakidetza.net)
Sergio Carrera (SERGIO.CARRERAREVILLA@osakidetza.net)
Marisol Arietaleanizbeaskoa (MARIASOLEDAD.ARIETALEANIZBEASCOASARABIA@osakidetza.net)
Veronica Arce (veronica.arce@osakidetza.net)
Nere Mendizabal (NERE.MANDIZABALGALLASTEGUI@osakidetza.net)
Anna Gine (ANNA.GINEMARCH@osakidetza.net)
Aitor Sanz-Guinea (Aitor.SanzGuinea@osakidetza.net)
Araceli Eskisabel (MARIAARACELI.ESKISABELGONI@osakidetza.net)
Ana Lopez (ANAMARIA.LOPEZRODRIGUEZ@osakidetza.net)
Rosa A martin (ROSAAMAIA.MARTINSANTIDRIAN@osakidetza.net)
Guillermo Lopez-Vivanco (GUILLERMO.LOPEZVIVANCO@osakidetza.net)
Gonzalo Grandes (GONZALO.GRANDESODRIOZOLA@osakidetza.net)

Version: 2
Date: 23 July 2014

Author's response to reviews: see over
Supervised physical exercise to improve the quality of life of cancer patients: the EFICANCER randomised controlled trial

Journal: BMC Cancer
Manuscript ID: 3004784231355022
Submitted: 10 July 2014

July 23, 2014

Dear Editorial board,

Please find attached the requested new version of the manuscript, revised according to your comments indicating the registration date in ClinicalTrials.gov and clarifying that this study was approved by the Ethics Committee of the Basque Country.

According to the requesting ethics and funding proofs:

(a) Ethical and Funding Approval Documentation:
I sent the requesting documents attached in an email to the following address BMCSeriesEditorial@biomedcentral.com. Documents: original informs of the Funding Approval, the Clinical Research Ethics Committee and the Primary Care Research Committee; and their own translations endorsed and signed by the director of our institution Dr. Gonzalo Grandes Odriozola

(b) Funding:
Our study is externally funded by the main Public Research Entity funding in Spain: Institute of Health Carlos III, ISCIII.

(c) Study status:
Our study is currently at the recruitment stage. We are still collecting data. 42 patients have enrolled up to date.

(d) Related articles:
We have not published any article related with this study yet

We very much appreciate your attention to our submission

Sincerely,

Aintzane Sancho
Sergio Carrera
Marisol Arieta Aleanizbeascoa
Verónica Arce
Nere Mendizabal Gallastegui
Anna Giné March
Aitor Sanz-Guinea
Araceli Eskisabel
Ana Lopez Rodriguez
Rosa Amaia Martín
Guillermo Lopez-Vivanco
Gonzalo Grandes